Author's response to reviews

Title: Interleukin-28B gene non-TT allele strongly predicts treatment failure for genotype 1 infected chronic hepatitis C patients with advanced fibrosis: a case control study

Authors:

Ching-Chih Hu Dr (nonahu5248@gmail.com)
Chih-Lang Lin Dr (wn49792000@yahoo.com.tw)
Liang-Che Chang Dr (lc2008@adm.cgmh.org.tw)
Cheng-Hung Chien Dr (cashtoto@gmail.com)
Li-Wei Chen Dr (leiwei@adm.cgmh.org.tw)
Ching-Jung Liu Dr (cjiuu@cgmh.org.tw)
Rong-Nan Chien Dr (ronald@adm.cgmh.org.tw)

Version: 4 Date: 1 December 2014

Author's response to reviews: see over
Dear chief editor,

We would like to submit our manuscript entitled “Interleukin-28B gene non-TT allele strongly predicts treatment failure for genotype 1 infected chronic hepatitis C patients with advanced fibrosis: a case control study” by Ching-Chih Hu, Chih-Lang Lin, Liang-Che Chang, Cheng-Hung Chien, Li-Wei Chen, Ching-Jung Liu, Rong-Nan Chien to “BMC Infectious Diseases” for publication as original article. This article demonstrated the strong influence of genetic factor on the therapeutic response in cirrhotic patients infected with hepatitis C virus.

All authors above have contributed significantly to this original article and are in agreement with the content of the manuscript. The study was performed in accordance with the ethical guidelines of the International Conference on Harmonisation for Good Clinical Practice. We certify that the results presented in this paper have not been published previously in whole or part, except in abstract form elsewhere and all authors have read and approved the final manuscript. In compliance with the Copyright Revision Act of 1976, effective Jan. 1, 1978, the authors transfers all copyright ownership of the manuscript to BMC Infectious Diseases in the event the material is published.

We have revised the manuscript by adding line numbering according to the Journal Editorial Office’s suggestion and resubmit the manuscript to BMC Infectious Diseases.

Ching-Chih Hu, M.D.
Liver Research Unit, Chang Gung Memorial Hospital and University,
Keelung, Taiwan
222, Mai-Chin Road, Keelung 20401, Taiwan
Telephone: 886-2-24313131 ext.3171
Fax: 886-2-24335342
E-mail address: nonahu5248@gmail.com